Author's response to reviews

Title: The prevalence of stunting, overweight and obesity, and metabolic disease risk in rural South African children

Authors:

Elizabeth W Kimani-Murage (lizmurage@gmail.com)
Kathleen Kahn (Kathleen.Kahn@wits.ac.za)
John M Pettifor (John.Pettifor@wits.ac.za)
Stephen M Tollman (Stephen.Tollman@wits.ac.za)
David B Dunger (dbd25@cam.ac.uk)
Xavier F Gómez-Olivé (xavier@agincourt.co.za)
Shane A Norris (san@global.co.za)

Version: 6 Date: 22 March 2010

Author's response to reviews:

Dear Editor,

The changes made include:

1. Corresponding author email has been added.

2. Boys, Girls in Row 1 of Tables 1 & 2 were centered.

Kind regards,

Elizabeth Kimani-Murage